Author's response to reviews

Title: Dietary Diallyl Disulfide supplementation attenuates ethanol-mediated pulmonary vitamin D speciate depletion in C57Bl/6 mice.

Authors:

  Michael L McCaskill (mmccaski@tulane.edu)
  Henry T Hottor (thottor@tulane.edu)
  Muna Sapkota (muna.sapkota@unmc.edu)
  Todd A. Wyatt (twyatt@unmc.edu)

Version: 2  Date: 29 September 2014

Author's response to reviews: see over
Dear Editors,

Thank you for considering to publish my most recent work. BMC Nutrition currently states an interest in publishing research concerning, but not limited to, public health nutrition, the biological underpinnings of nutrition in the body, and clinical nutrition. This submitted manuscript illustrates that daily exposure of dietary diallyl disulfide can ameliorate chronic ethanol-mediated reductions in serum and pulmonary vitamin D levels. I believe that the submitted manuscript accompanying this cover letter which describes dietary mediators of vitamin D and pulmonary antimicrobial peptides is within the scope and aim of your fine journal. The modulation of the vitamin D pathway has recently been rediscovered as a potential cause, and solution of several illnesses including Chronic Obstructive Pulmonary Disorder, Multiple Sclerosis, Acute Respiratory Distress Syndrome, and Sepsis. I hope I have the honor of having my research published in BMC Nutrition. If you have any questions or concerns, please do not hesitate to contact me. Thank You for your consideration.

Revision: Line Numbers have been added to each line of the manuscript.

Best Regards,

Michael McCaskill Ph.D., MPH
Assistant Professor
**Peter Bergman**, MD, PhD, Docent, Associate Professor, Group Leader

**D A Hughes*** and **R Norton†**
*Institute of Food Research, Norwich Research Park, University of East Anglia, Norwich, UK
†School of Pharmacy, University of East Anglia, Norwich, UK
Correspondence: D. A. Hughes, Institute of Food Research, Norwich Research Park, Norwich NR4 7UA, UK. E-mail:davidA.hughes@bbsrc.ac.uk

**Author information ▼ Article notes**

*Nutrition & Metabolism* publishes content focused on the integration of nutrition, exercise physiology, clinical investigations, and molecular and cellular biochemistry of metabolism. The areas of interest of *Nutrition & Metabolism* encompass studies in obesity, diabetes, lipidemias, metabolic syndrome and exercise physiology that have an underlying basis in metabolism. Likewise, we seek submission of manuscripts on the biochemistry of metabolism, cell signaling, molecular and cellular biology of nutrients, nutrient gene interactions and other areas that have implications for human nutrition and medicine. Current, but not exclusive, interests are metabolic effects of diet composition, interactions of macronutrients, effect of nutrients on gene expression, metabolic control and compartment models, nutritional effect of hormones, and genomic analysis of dietary phenomena.